Author’s response to reviews

Title: Implication of metastasis suppressor gene, Kiss-1 and its receptor Kiss-1R in colorectal cancer

Authors:

Ke Ji Dr. (jk1115@hotmail.com)
Lin Ye Dr. (Yel@cardiff.ac.uk)
Rachel Hargest Dr. (Hargest@cardiff.ac.uk)
Malcolm D Mason Prof. (MasonMD@cardiff.ac.uk)
Wen G Jiang Prof. (JiangW@cardiff.ac.uk)
Fiona Ruge Mrs (ruge@cardiff.ac.uk)

Version: 5 Date: 17 September 2014

Author’s response to reviews:

RESPONSE TO EDITORIAL COMMENTS

We wish to thank the editor and the editorial team for the suggestion and comments. The following is the information requested.

-----------------------------------

Please include an authors’ contributions section after competing interest.

In order to give appropriate credit to each author of a paper, the individual contributions of authors to the manuscript should be specified in this section.

-----------------------------------

We have provided the required information which can be seen in lines 362-368 in the revised version and reads as follows:

'Author’s contributions:

KJ and LY carried out the expression analysis, cell and molecular based work and data analyses. FR carried out immunohistochemical and histological analyses. RH participated in sample collection and patient followup. MDM participated in study design. WGJ participated in expression analysis, gene knockdown design and data analyses. All participated manuscript preparation and approved the final manuscript.'